Author’s response to reviews

Title: "Self-development Matters" - Perception of Sakhis (CHWs) assessing self-development outcomes of their participation in the HBNC Program

Authors:
Anagha Mahajani (anagha.mahajani.ext@ambujacement.com)
Abhijit Prabhughate (abhijit.prabhughate@ambujacement.com)
Pearl Tiwari (pearl.tiwari@ambujacement.com)
Shubhangi Sohoni (shubhangi.sohoni.ext@ambujacement.com)
Ajay Phatak (ajaygp@gmail.com)
Vallaree Morgaonkar (vallarimorgaonkar@gmail.com)
Somashekhar Nimbalkar (somu_somu@yahoo.com)

Version: 2 Date: 14 Jul 2017

Author’s response to reviews:

Please upload all figures as separate figure files and remove them from the main manuscript.

This has been done with three separate word documents being uploaded.

Please do not include the figure captions within the figure files, but instead place them at the end of the main manuscript in a section titled “Figure legends”

This has been done